Author’s response to reviews

Title: The global switch from trivalent oral polio vaccine (tOPV) to bivalent oral polio vaccine (bOPV): Facts, Experiences and Lessons Learned from the south-south zone; Nigeria, April 2016

Authors:

Enya Bassey Bassey (bassey69@yahoo.com; bbenya69@gmail.com)

Fiona Braka (brakaf@who.int)

Rui Gama Vaz (ruivaz@who.int)

William Komakech (komakech@who.int)

Sisay Gashu Tegegne (gashutegegns@who.int)

Sylvester Toritseju Maleghemi (maleghemis@who.int)

Richard I Koko (kokor@who.int)

Thompson U Igbo (igbut@who.int)

Faith Ireye (ireyef@who.int)

Sylvester O Agwai (agwais@who.int)

Godwin Ubong Akpan (akpang@who.int)

Abdul-Aziz Garba Mohammed (mohammedab@who.int)

Angela Okocha-Ejeko (peaches_spears@rocketmail.com)

Version: 3 Date: 10 Jul 2017

Author’s response to reviews:

10th July 2017.Filippo Ansaldi,BioMed Central (BMC) Infectious Diseases.Dear Sir,In response to the reviewer’s comment on the manuscript titled ‘The global switch from trivalent Oral Polio Vaccine (tOPV) to bivalent Oral Polio Vaccine (bOPV): Facts, experiences and Lessons Learned from the south-south zone; Nigeria, April 2016’, the authors have made the recommended changes (in red in text) outlined below:1.An update has been made to reflect the confirmed cVDPV2 cases in the Democratic Republic of Congo as recommended. See text in red in the ‘discussion’ session 2.General reorganization of the whole manuscript has also been done to hopefully eliminate all grammatical errors for an easier read. See changes in red in text. Thank you very much for your
kind consideration and I look forward to hearing back from the journal soonest. Warm Regards, Dr Bassey Enya Bassey